Author's response to reviews

Title: Plasma concentrations of soluble cytokine receptors in euthymic bipolar patients with and without subsyndromal symptoms

Authors:

Turan Cetin (turancetin34@hotmail.com)
Sinan Guloksuz (sguloksuz@yahoo.com)
Esin Aktas Cetin (esinaktas@yahoo.com)
Sema Bilgic Gazioglu (sbilgic@istanbul.edu.tr)
Gunnur Deniz (gdeniz@istanbul.edu.tr)
Timucin E Oral (etoral@superonline.com)
Jim van Os (j.vanos@maastrichtuniversity.nl)

Version: 6 Date: 4 September 2012

Author's response to reviews:

Dear Editor,

Many thanks for allowing us to send in a revised version of this manuscript. Please find attached the revised manuscript. We have addressed minor modifications point by point, as requested.

1) Drop phrase referencing "ongoing pro-inflammatory process"
Thank you for this comment. We have dropped the whole sentence